Author's response to reviews

Title: Correlation between delayed-enhancement magnetic resonance and nitrates myocardial Tc-99m tetrofosmin scintigraphy in myocardial infarction: a case report

Authors:

   Mauro Feola (m_feola@virgilio.it)
   Gian Luca Rosso (rossogl@tiscali.it)
   Alberto Biggi (biggi.a@ospedale.cuneo.it)
   Stephane Chauvie (chauvie.s@ospedale.cuneo.it)
   Giovanni Leonardi (leonardi.g@ospedale.cuneo.it)
   Franca Margaria (margaria.f@ospedale.cuneo.it)
   Maurizio Grosso (grosso.m@ospedale.cuneo.it)
   Valeria Ferrero (valeria.ferrero@hotmail.com)

Version: 4 Date: 19 August 2007

Author's response to reviews:

To the reviewer Prof. Ribichini:

Description of ECG: modified in Fig.1.

Laboratory data .....: added in page 2.

Line 14 of the text: changed minute in minutes.

Last sentence....: changed in page 3.

Comment about OAT study: reworded in page 3.

Finally we added "oral" before postnitrate myocardial SPECT.

To the reviewer Dr. Lombardi:

We underline the possible advantages of MRI-DE in comparison to myocardial SPECT in page 1 and changed, as advised, the term "fibrosis" with "necrosis".

I would like to thank the two reviewers that improved the quality of the manuscript.